Title: Haemorrhagia post partum; an implementation study on the evidence-based guideline of the Dutch Society of Obstetrics and Gynaecology (NVOG) and the MOET (Managing Obstetric Emergencies and Trauma-course) instructions; the Fluxim study

Authors:

Mallory D Woiski (M.Woiski@obgyn.umcn.nl)
Rosella PMG Hermens (R.Hermens@iq.umcn.nl)
Johanna M Middeldorp (j.m.middeldorp@lumc.nl)
Jan A Kremer (J.Kremer@obgyn.umcn.nl)
Marco A Marcus (m.marcus@mumc.nl)
Maurice GAJ Wouters (mja.wouters@vumc.nl)
Richard P Grol (R.Grol@iq.umcn.nl)
Fred K Lotgering (F.Lotgering@obgyn.umcn.nl)
Hubertina CJ Scheepers (hcj.scheepers@mumc.nl)

Version: 7 Date: 24 January 2010

Author's response to reviews: see over
Nijmegen, 24 January 2010

Subject: Study protocol Haemorrhagia post partum; an implementation study on the evidence-based guideline of the Dutch Society of Obstetrics and Gynaecology (NVOG) and the MOET (Management of Obstetric Emergencies and Trauma-course) instructions; the Fluxim study

Dear Editor,

We are very content with the acceptance of the manuscript for publication. In the second page you will find the point by point description of the changes made.

Yours sincerely,

Mallory Woiski, MD, corresponding author

Radboud University Nijmegen
Department of Obstetrics and Gynaecology

Huispostnummer 791
P.O.Box 9101
6500 HB Nijmegen

The Netherlands

Tel: +31 2436 14747
Email: M.Woiski@obgyn.umcn.nl
The changes made

1. The structure has been adjusted like recommended into the suggested headings.
2. The text ‘study protocol’ has been removed
3. The address of the correspondence has been corrected by adding ‘Radboud University Nijmegen’
4. All the underlining has been changed into bold
5. The tracking has been removed
6. The abbreviations has been changed into a paragraph
7. The references has been updated with correction of references 1 and 5 (corrected to ‘data file type’ of reference).
8. Reference 4 has been replaced by a published article by the same author
9. Reference 6 has been changed like suggested by the reference recommendation site of the BMC
10. Reference 3 and 9 were the same and has been changed into one.
11. The manuscript has been checked on typographical errors.
12. Additional files has been removed.